Author's response to reviews

Title: B-vitamin consumption and the prevalence of diabetes and obesity among the US adults: population based ecological study

Authors:

Shi-Sheng Zhou (zhouss@ymail.com)
Da Li (dlulida@qq.com)
Yi-Ming Zhou (zhouym@nips.ac.jp)
Wu-Ping Sun (sunwupings@yahoo.com.cn)
Qi-Gui Liu (liuqigu1@163.com)

Version: 7 Date: 2 December 2010

Author's response to reviews:

Dear Editorial team,

The following is a list of the changes made to the revised version:

1) Page 18, in the last sentence: “Figure 6 C” was changed to “Figure 4E”, and “a decreasing trend in the per capita sugar consumption” was changed to “a decreasing trend in sugar contribution to per capita carbohydrate consumption”.

2) Page 39, the title of Figure 9 “The relationships among the consumption of the main nutrients in the US” was changed to “Relationships between per capita B-vitamin consumption and energy and protein consumption in the US”

3) The quality of each figure was improved.